Author's response to reviews

Title: The radiosensitizing effects of doranidazole on human colorectal cancer cells exposed to high dose irradiation

Authors:

Li Zhang (tyouri19652004@yahoo.com.cn)
Aimin Gong (yinke760109@163.com)
Jun Ji (cancer-1976@hotmail.com)
Yuanyuan Wu (wuyuanyuan9@yahoo.com.cn)
Xiaoyu Zhu (zhuxiaoyu66@yahoo.com.cn)
Suqing Lv (lusuqing341@hotmail.com)
Honazhu Lv (pole163@163.com)
Xizhuo Sun (xizhuomd@163.com)

Version: 4 Date: 20 August 2007

Author's response to reviews: see over
Dear Editor

Our manuscript, “The radiosensitizing effect of Doranidazole on human colorectal cancer cells exposed to high dose irradiation” by Li Zhang, Aimin Gong, Jun Ji, Yuanyuan Wu, et al. was re-reviewed by re-reviewer for publication in “BMCCancer”. We have corrected it as the re-reviewer indicates. (As below):

Reviewer # 1 (Halil Kavgaci)
Revise for reviewer’s comments
Reviewer’s report
Title: The radiosensitizing effects of doranidazole on human colorectal cancer cells exposed to high dose irradiation
Version: 3 Date: 9 August 2007
Reviewer: Halil Kavgaci
Reviewer’s report:
"The radiosensitizing effects of doranidazole on human colorectal cancer cells exposed to high dose irradiation" titled revised manuscript evaluated.
My advice for this manuscript:

Table 1 first line irradiation will be added.

We have corrected as the re-reviewer indicates. (See the table 1)

Table 2 "Effect of Doranidazole and anticancer agents on the cells". Which cell or cells will be mentioned.
This paper can be published after the corrections are made.

We have corrected as the re-reviewer indicates. (See the table 2)

Sincerely yours,

20, Aug 2007

Li Zhang M.D., Ph.D
Department of Central Laboratory, Dalian Municipal Central Hospital, Xinan Road 826, Shahekou-qu, Dalian, 116033, China
Phone: + 86-411-84441341, FAX: + 86-411-84441341
E-mail address: tyouri19652004@yahoo.com.cn